Author’s response to reviews

Title: The management of children with bronchiolitis in the Australasian hospital setting: Development of a clinical practice guideline. A PREDICT project.

Authors:

Sharon O'Brien (sharon.obrien@health.wa.gov.au)
Sally Wilson (Sally.Wilson@curtin.edu.au)
Fenella Gill (fenella.gill@health.gov.wa.au)
Elizabeth Cotterell (liz.cotterell@une.edu.au)
Meredith Borland (meredith.borland@health.gov.wa.au)
Edward Oakley (ed.oakley@rch.org.au)
Stuart Dalziel (sdalziel@adhb.govt.nz)

Version: 2 Date: 23 Nov 2017

Author’s response to reviews:

Changes as requested have been completed -

1. A "Declarations" section has been included at the end of the manuscript. It contains a list of abbreviations, as well as sections on 'Ethics approval and consent to participate', 'consent for publication', 'Availability of data and materials', 'Competing interests', 'Funding', 'Authors' Contributions' and 'Acknowledgements'.

2. Manuscript section sub title has been changed to Data collection, quality, and reporting on page 10.

3. A section titled “Supplementary files” has been added.